Author's response to reviews

Title: Social support and age influence distress outcomes differentially across urban-regional and remote Australia

Authors:

Joanne Allen (Joanne.Allen@newcastle.edu.au)
Kerry J Inder (Kerry.Inder@newcastle.edu.au)
Terry J Lewin (Terry.Lewin@hnehealth.nsw.gov.au)
John Attia (John.Attia@newcastle.edu.au)
Brian J Kelly (Brian.Kelly@newcastle.edu.au)

Version: 2 Date: 20 February 2012

Author's response to reviews:

Corrected a minor formatting error in manuscript and re-numbered Tables and Figures as the auto-numbering system had labelled all as Figure/Table '1' when they were separated into individual files.